Author’s response to reviews

Title: Multicentric, multifocal, and recurrent osteoid osteoma of the hip: first case report

Authors:

Maria Cristina Cortese (cristinacortese1010@gmail.com)
Domenico Albano (albanodomenico@me.com)
Carmelo Messina (carmelomessina.md@gmail.com)
Giuseppe Perrucchini (pergiudoc@yahoo.com)
Enrico Gallazzi (enrico.gallazzi@gmail.com)
Mauro Battista Gallazzi (maurobattista.gallazzi@gpini.it)
Primo Andrea Daolio (primoandrea.daolio@gpini.it)
Luca Maria Sconfienza (io@lucasconfienza.it)

Version: 3 Date: 19 Mar 2019

Author’s response to reviews:

Dear Editor-in-Chief,

Thanks for the opportunity to resubmit our paper #BMSD-D-19-00066R1 entitled "Multicentric, multifocal, and recurrent osteoid osteoma of the hip: first case report" to be considered for publication in BMC Musculoskeletal Disorders. We have read the Editor’s comments and we believe that we have responded substantively to all of them.

Specifically

Editor:

Comment 1: The authors have responded to all the queries raised by the reviewers. I am pleased to inform you that your manuscript is potentially acceptable.

However, as one of the reviewers have also suggested, the manuscript needs some language corrections
by a native English speaker or English editing service before being published.

If the manuscript was sent to a language editing company prior to submission please indicate it in the Acknowledgement section.

Answer 1: We thank the Editor for his comment. As suggested, we have sent the manuscript to a native English speaker for English editing.

The Authors